

Article



Scheduling and Power Control for Wireless Multicast Systems via Deep Reinforcement Learning [†]

Ramkumar Raghu¹, Mahadesh Panju¹, Vaneet Aggarwal^{2,*} and Vinod Sharma¹

- ¹ Indian Institute of Science, Karnataka 560012, India; ramkumar@iisc.ac.in (R.R.); mahadesh@iisc.ac.in (M.P.); vinod@iisc.ac.in (V.S.)
- ² School of Industrial Engineering and School of Electrical and Computer Engineering, Purdue University, West Lafayette, IN 47907, USA
- * Correspondence: vaneet@purdue.edu
- + Preliminary version of a part of this paper was presented in 57th Annual Allerton Conference on Communication, Control, and Computing, 2019.

Abstract: Multicasting in wireless systems is a natural way to exploit the redundancy in user requests in a content centric network. Power control and optimal scheduling can significantly improve the wireless multicast network's performance under fading. However, the model-based approaches for power control and scheduling studied earlier are not scalable to large state spaces or changing system dynamics. In this paper, we use deep reinforcement learning, where we use function approximation of the Q-function via a deep neural network to obtain a power control policy that matches the optimal policy for a small network. We show that power control policy can be learned for reasonably large systems via this approach. Further, we use multi-timescale stochastic optimization to maintain the average power constraint. We demonstrate that a slight modification of the learning algorithm allows tracking of time varying system statistics. Finally, we extend the multi-time scale approach to simultaneously learn the optimal queuing strategy along with power control. We demonstrate the scalability, tracking and cross-layer optimization capabilities of our algorithms via simulations. The proposed multi-time scale approach can be used in general large state-space dynamical systems with multiple objectives and constraints, and may be of independent interest.

Keywords: multicasting; scheduling; queuing; deep reinforcement learning; quality of service; power control; dynamics tracking; multi-timescale stochastic optimization

1. Introduction

Content services, such as Netflix, Prime Video, etc., have dramatically increased the demand for high-definition videos over mobile networks. Almost 78% of mobile data traffic is expected to be due to these mobile videos [1]. It is observed that the request traffic for these contents have multiple redundant requests [2]. Next generation wireless networks are being constantly upgraded to satisfy these exploding demands by exploiting the nature of the request traffic. Serving the redundant requests simultaneously is a natural way to utilize network resources efficiently. Thus, efficient multicasting is studied widely in the wireless networking community.

A multicast queue with network coding is studied in [3] with an infinite library of files. The case of slotted broadcast systems with one server transmitting to multiple users is studied in [4]. Some recent works [5] use coded caching to achieve multicast. This approach uses local information in the user caches to decode the coded transmission and provides improvement in throughput by increasing the effective number of files transferred per transmission. This throughput may get reduced in a practical scenario, due to queuing delays at the base station/server. Ref. [6] addresses these issues, analyzes the queuing delays and compares it with an alternate coded scheme with LRU caches (CDLS), which provide improvement over the coded schemes in [5]. A more recent work in



Citation: Raghu, R.; Panju, M.; Aggarwal, V.; Sharma, V. Scheduling and Power Control for Wireless Multicast Systems via Deep Reinforcement Learning. *Entropy* **2021**, *23*, 1555. https://doi.org/ 10.3390/e23121555

Academic Editor: Raúl Alcaraz

Received: 12 October 2021 Accepted: 18 November 2021 Published: 23 November 2021

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2021 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https://creativecommons.org/ licenses/by/4.0/). this direction, Ref. [7] provides alternate multicast schemes and analyzes queuing delays for such multicast systems. In [7], it is shown that a simple multicast scheme can have significant gains over the schemes in [5,6] in a high-traffic regime.

We further study the multicast scheme proposed in [7] in this paper. This multicast queue merges the requests for a given file from different users, arriving during the waiting time of the initial requests. The merged requests are then served simultaneously. The gains achieved by this simple multicast scheme, however, are quickly lost in wireless channels, due to fading. It suffers from users with bad channels, thereby decreasing the QoS, even for users with good channels. In [7], we studied this problem and proposed novel schemes, which provide significant multicast gains under fading as compared to the simple multicast. In [7], we also theoretically analyzed these queuing schemes, showed that our queues are always stable and provided approximate theoretical expressions for mean sojourn times. Further, in [8], we showed that state-dependent power control under an average power constraint can significantly improve the average delays experienced by users.

The queuing schemes and the power control policy proposed in [8], though they provide improved delays, have the following limitations. (1) The queuing scheme which performs best depends on the system parameters, such as the size of the system, the request rate, etc. (2) The algorithm to obtain the power control policy is not scalable with the number of users and the number of states of the channel gains. Additionally, the policy does not adapt to changing the system statistics, which in turn, depends on the power control policy. (3) The queuing schemes and power control are dealt with individually. This paper tries to overcome the above limitations of the scheme in [8].

We first provide algorithms for the two optimization problems individually and then combine the two algorithms to obtain the overall optimal queuing strategy and the power control. Stochastic optimization ([9]) is a useful tool to obtain the optimally parameterized queuing strategy. However, for the convergence of stochastic optimization algorithms, a careful approximation of stochastic gradients is necessary. One challenge here is that the cost to be optimized is the mean stationary sojourn time of the requests to be delivered. We propose a new deep assisted gradient approximation algorithm, where the novelty is in deriving the gradients from a deep network assisted by a memory. This memory helps retain the history of the explored regions and also allows adaptation to changing system dynamics in an online fashion. The replay memory and online training of the deep network adds an important feature called importance sampling to the stochastic optimization, which improves the confidence (lower variance) in the gradient descent steps.

Multicast systems with power control can be conveniently modeled as a Markov decision process (MDP) but with large state and action spaces. Obtaining transition probabilities and the optimal policy, however, for such large MDPs is not feasible. Reinforcement learning, particularly deep reinforcement learning [10], is a natural tool to address such problems. Reinforcement learning can be used even when the transition probabilities are not available. However, a large state/action space can still be an issue. Using function approximation via deep neural networks can provide significant gains. Several deep reinforcement learning techniques, such as deep Q-network [11], trust region policy optimization (TRPO) [12], proximal policy gradient (PPO) [13], etc., have been successfully applied to several large state-space dynamical systems, such as Atari [14], AlphaGo [15], etc. DQN is based on value iteration. TRPO and PPO are policy-gradient-based methods. Policy–gradient methods often suffer from high variance in sample estimates and poor sample efficiency [10]. Value-iteration-based deep RL methods, such as DQN, have been theoretically shown to have better performance [16], due to target network and replay memory and providing a global minimum.

We propose a constrained optimization variant of DQN based on multi-timescale stochastic gradient descent [9] for power control, which can track the system statistics. Finally, we develop an algorithm which combines the above two algorithms to obtain an optimal queuing strategy and power control policy.

The major contributions of this paper are as follows:

- A novel deep assisted stochastic gradient descent (DSGD) algorithm for obtaining the best queuing strategy from a given set.
- Proposing two modifications to DQN to accommodate constraints and system adaptations. The constraints can be met by using a Lagrange multiplier. The appropriate Lagrange multiplier is also learned via a two-timescale stochastic gradient descent. We call this algorithm adaptive constrained DQN (AC-DQN).
- Unlike DQN, AC-DQN can be applied to the multicast systems with constraints, as in [8], to learn the power control policy, online. The proposed method meets the average power constraint while achieving the global optima as achieved by the static policy proposed in [8] for a small-scale setup of the problem.
- We demonstrate the scalability of our algorithms with the system size (number of users, arrival rate, complex fading).
- Finally, using the above two algorithms, we propose a generalized algorithm called integrated DSGD and AC-DQN (IDA) to optimize systems with multiple objectives and constraints. Particularly, this algorithm is useful in any wireless network with cross-layer objectives, such as ours. IDA is a three-timescale stochastic optimization algorithm for obtaining both the queuing strategy (unconstrained network layer objective) and power control (constrained physical layer objective), simultaneously.
- We also show that AC-DQN and IDA can track the changes in the dynamics of a non-stationary system, e.g., change of arrival rate or number of users over the time of a day, and achieve optimal performance.

We show via simulations that our algorithms choose the optimal policy among the given set of policies. Additionally, the power control policy obtained via our algorithm improves the delay performance of the multicast network by more than 50%, compared to the constant power policy. Our algorithms work equally well when we replace DQN with its improvements, such as DDQN [17]. In fact we ran our simulations with the DDQN variant of AC-DQN and achieved similar performance. It is worth noting that, even though we demonstrate the power of deep (reinforcement) learning, in improving schemes in [7,8], the proposed deep algorithms themselves are generic and can be applied to any dynamical system with multiple objectives, constraints and large state spaces.

Related Works

Queuing and Power control in Multicast Systems: Multicast queuing and scheduling is studied in [3,18–20]. The works in [3,18,19] propose schemes for network-coded multicast systems and analyze the stability of the proposed multicast queues. Unlike these works, we use, as in our previous work [7,8], a simple uncoded multicast queue, which is always stable. In [7], we show that our queuing schemes perform much better than the coded multicast schemes in high traffic regimes. In the current work, we improve the results in [7,8] by providing novel deep-learning-based queuing strategies. Ref. [20] proposes a multicast scheduling scheme for Poisson traffic. However, there is no power control, and the proposed queue is not always stable. The effect of multicasting and caching on energy cost in a delay-tolerant content-centric network is studied in [21]. The work, however, does not consider the effect of the queuing delay considered in the paper, and does not have any constraints on the transmit power. Power control in multicast systems is studied in [22–24]. In [22], power allocation optimizes the ergodic capacity while maintaining certain minimum rate requirements at the users and average power constraints. In [23], the authors minimize a utility function via linear programming under SINR constraints at the users and transmit power constraints at the transmitter. Refs. [22,23] derive an optimal power control policy for delivery to *all* the users, whereas this paper considers delivery to a random subset of users requesting a file at that time. Additionally, the power control policies in [22,23] require knowledge of system statistics and are not scalable for our system. Ref. [24] considers MDP-based scheduling and power control in content-centric multicast systems. The work in [24] uses fixed channel states, requires statistics of queue state transitions and does not have any constraint on the average transmit power. Further, the state dimension of the system increases with both the number of users and files, whereas in our work, the dependence is only on the number of users. Thus, compared to the above-mentioned works, our scheme is more practical, computationally scalable, does not require knowledge of system statistics (traffic intensity, and fading distributions) and can track changing system statistics.

Deep Learning in Wireless Multicast systems: The ability of DeepRL to handle large state-space dynamic systems is being exploited in various multicast wireless systems/networks. In [25], the authors study a resource allocation problem in unicast and broadcast transmissions. The DeepRL agent learns and selects the power and frequency for each channel to improve the rate under some latency constraints. Like in our work, they also introduce constraints via Lagrange multipliers. However, the Lagrange multiplier is constant, and the agent does not learn it. Thus, the agent also does not adapt if the system dynamics changes, as the Lagrange constant is fixed and the learning rate decays with time. To obtain the appropriate Lagrange multiplier is computationally expensive and requires known system statistics. Another work, Ref. [26], applies unconstrained deep reinforcement learning to multiple transmitters for a proportionally fair scheduling policy by adjusting individual transmit powers. Ref. [27] applies DeepRL in queuing in a coded caching-based multicast system, which is shown to be inferior to our multicast schemes in high traffic rate regions. For more literature on deep learning applications to wireless multicast systems, see the detailed survey in [28].

For a detailed exposition on constrained MDPs, see [29]. Some recent works on reinforcement learning provide convergence guarantees for the tabular model-free Q-learning, using the minimax approach [30], model-based online policy optimization approach [31], tabular model-based Q-learning approach [32], and tabular primal-dual approach [33]. The approaches in these works are not demonstrated on large state spaces, in part due to the increased complexity of the tabular algorithm in [30,32,33], and linear MDP assumption in [31]. Ref. [34] introduces constrained reinforcement learning based on TRPO. Unlike ours, the work considers discounted constraints. In [35], a Lagrange-based actor-critic approach for constrained RL is proposed. Since [31,34,35] are policy-based approaches, they suffer from high variance when multiple evaluations are unfeasible. In [36], an alternate approach with two value functions for reward and constraint (cost) with an actor-critic policy update is proposed. Here, at each step, a convex relaxation-based optimization is used to obtain the optimal parameter of value functions. We note that the convex optimization step at each iteration is computationally more intensive than a simple SGD step. Thus, the above-mentioned policy iteration methods either have high variance in practical systems or are computationally intensive. These issues make it difficult to track the changing dynamics in practical systems, as we can in our case. To the best of our knowledge, ours is the first constrained value iteration-based deep RL algorithm for constrained MDPs. The use of replay memory and a target network helps reduce the estimator variance in our algorithm. These features also increase the practical applicability of our algorithm.

The rest of the paper is organized as follows. Section 2 explains the system model and motivates the problem. Section 3 presents our deep-learning-based optimal queuing algorithm. Section 4 motivates the power control problem and briefly explains the power control algorithm proposed in [8]. Section 5 presents the proposed DeepRL algorithm AC-DQN for scalable, improved power control. Section 6 presents our novel deep multitimescale algorithm to achieve scalable cross-layer optimization of queuing and power control and provides optimal performance for the multicast system. Section 7 demonstrates our algorithms via simulations, and Section 8 concludes the paper.

2. System Model

We consider a system with one server transmitting files from a fixed finite library to a set of users (Figure 1). We denote the set of users by $\mathcal{L} = \{1, 2, \dots, L\}$ and the set of files by $\mathcal{M} = \{1, 2, \dots, M\}$. The request process for file *i* from user *j* is a Poisson process of rate λ_{ij} , which is independent of the request processes of other files from user *j* and also from

other users. The total arrival rate is $\lambda = \sum_{i,j} \lambda_{ij}$. The requests of a file from each user are queued at the server until the user successfully receives the file. All the files are of length *F* bits. The server transmits at a fixed rate, *R* bits/s. Thus, the transmission time for each file is T = F/R.

The channels between the server and the users experience time varying fading. The channel gain of each user is assumed to be constant during the transmission of a file. The channel gain for the *j*th user at the *t*th transmission is represented by $H_j(t)$. Each $H_j(t)$ takes values in a finite set and forms an independent identically distributed (i.i.d) sequence in time, as in [37]. The channel gains of different users are independent of each other and may have different distributions. Let $H = (H_1, \dots, H_L)$.

Since the requests from the users are queued at the server, every request awaits its turn for transmission, and thus experiences a queuing delay, which is random in nature. The distribution of this random delay depends on the queuing policy. Additionally, unsuccessful transmissions due to fading adds further delay, experienced by each request. We denote by random variable *D* the overall delay experienced by each request due to both queuing and fading. If t_A is the time of arrival of a request to the server and t_S is the time instance representing the end of successful transmission/service of the request, then the random delay/sojourn time *D* is given by $D = t_S - t_R$. Further, E[D] denotes the stationary mean sojourn time experienced by each request.



Figure 1. System model.

More details of the system are described in the following sections as follows. Section 2.1 describes the basic multicast queue proposed in [7]. The queuing schemes to mitigate the effects of fading studied in [8] are also presented. Section 2.2 parameterizes the queuing schemes. Section 3 provides an online learning scheme to obtain the optimal policy for a given setup. In Sections 4.1 and 4.2, we summarize the results from [8], which show that using power control can further improve the performance and the algorithm used to obtain the optimal power policy. We will see that this algorithm is not scalable. Then in Section 4.3, we provide the MDP of the power control problem. In Section 5, we present the scalable DeepRL solution for this formulation.

2.1. Multicast Queue

For scheduling transmissions at the server, we consider the *multicast queue* studied in [8]. In this system, the requests for different files from different users are queued in a single queue, called the multicast queue. In this queue, the requests for file *i* from all users are merged and considered a single request. The requested file and the users requesting it is denoted by (i, \mathbb{L}_i) . In other words, \mathbb{L}_i is the list of users interested in file *i*. A new request for file *i* from user *j* is merged with the corresponding entry \mathbb{L}_i if it already exists. Otherwise, it is appended to the tail of the queue. The service/transmission of file *i* serves all the users in \mathbb{L}_i , possibly with errors, due to channel fading.

The random subset of users served by the multicast queue at the *t*th transmission is denoted by the random binary vector, $V(t) = (V_1(t), \dots, V_L(t))$, where $V_i(t) = 1$

It is shown in [7] that the above multicast queue performs much better than the multicast queues proposed in the literature before. *The main difference compared to previous multicast schemes is that in this scheme, all requests of all the users for a given file are merged together over time.* One direct consequence of this is that the queue length at the base station does not exceed M. Thus, the delay is bounded for all traffic rates. It is worth noting that this is unique to our queues in [7] and none of the queues proposed in the literature have this feature. In fact, the mean delays are often better than the coded caching schemes proposed in the literature as well for most of the traffic conditions.

In a fading scenario, where the different users have independent fading, the performance of this scheme can significantly deteriorate because of the multiple retransmissions required to successfully transmit to all the users needed. Thus, in [8], multiple queuing strategies are proposed and compared to recover the performance of the system and reduce the mean delay substantially. Some of these are also fair to different users in the sense that the users with good channel gains do not suffer because of users with bad channel gains. We comment more on this in the following. We now briefly present the schemes proposed in [7,8] for clarity.

Retransmit: This is the simplest scheme proposed in [7]. Here, the multicast queue is serviced from head to tail. The head of the line is retransmitted until all the users in it are serviced. The new requests are added to the queue in a similar manner to the simple multicast. This naive scheme works very well in a low request rate regime; however, it performs poorly for high request rates and severely deteriorates the delays experienced by users with good channels.

Single queue with loop-back (1-LB): The multicast queue is serviced from head to tail. When a file is transmitted, some of the users receive the file successfully and some users may receive the file with errors. In the case of unsuccessful reception by some users, the file is retransmitted. A maximum of N ($1 \le N \le \infty$) transmission attempts are made. If there are some users who have not received the file within N transmission attempts, the request (tuple (i, \mathbb{L}_i) with \mathbb{L}_i , now modified to contain only the set of users who have not received the file i successfully) is fed back to the queue. If there is another pending request in the queue for the same file (a request for the file which came during the current transmission), it is merged with the existing request. Otherwise, a new request for the same file with unsuccessful users is inserted at the tail of the queue.

Defer queue with loop back (2-LB): This strategy has two queues for servicing the requests: a multicast queue and a defer queue. The multicast queue is similar to the queue mentioned in the beginning of this section and is serviced from head to tail. The defer queue is an additional queue to handle unsuccessful transmissions as follows. When a file is transmitted, some users may receive the file with errors. In the case of unsuccessful reception by some users after a maximum of *N* transmissions, the file request and the unserviced users are queued in the defer queue. Such requests stay in the defer queue until a new request for the same file arrives. On the arrival of the new request, the new request is merged with the older requests in the defer queue and moved to the tail of the multicast queue. If no such old requests exist in the defer queue, the new request is merged/added to the multicast queue. This queue is shown to provide lower delay to good channel users than to bad channel users.

The performance of each of these queues depends on the system parameters, transmission power policy, arrival rate, etc. If the channel gain statistics of different users are different, say, one group with good statistics and another with bad statistics, then the rate of transmission R and N can decide on the preference one is giving to the two groups of users. A higher R and lower N will give more preference to the good users at the cost of the bad users. For simplicity of presentation, we consider the case of N = 1 for all the queuing strategies in this paper.

2.2. Parametrization of Queueing Strategies

To adaptively optimize the queuing strategy according to the system parameters, it is convenient to first parameterize them. We propose a simple parametrization, using probabilities for each queuing strategy. That is, at the end of every service instance, if some users have not received the file successfully, the multicast queue chooses to retransmit the head of the line (HoL) request with probability p_1 , loopback HoL with probability p_2 , or defer HoL with probability p_3 , such that $\sum_{j=1}^{j=3} p_j = 1$. Thus, $\overline{p} = [p_1, p_2, p_3]$ parameterizes the queuing strategy. Here, $\overline{p} \in \mathbb{P}$, where \mathbb{P} is the probability simplex, $\mathbb{P} = \{[p_1, p_2, p_3] \in [0, 1]^3 : \sum_{j=1}^{j=3} p_j = 1\}$. Observe that $\overline{p} = [1, 0, 0]$, [0, 1, 0], and [0, 0, 1] represent retransmit, loopback, and defer strategies. In the next section, we provide an algorithm to obtain the optimal \overline{p} .

3. Deep Learning for Optimal Queueing

We are interested in finding the optimal \overline{p} among the parameterized queuing strategies in Section 2.2 that gives the least average delay. From our previous work (Proposition 1, [7]), it can be shown that for any parameter \overline{p} , there exists a stationary mean sojourn time, $E_{\overline{p}}[D]$, where D is the sojourn time and E is the expectation. In this section, we propose an online deep learning algorithm to learn $\overline{p}^* = \underset{\overline{p} \in \mathbb{P}}{\operatorname{argmin}} E_{\overline{p}}[D]$. However, the map

 $f: \overline{p} \mapsto E_{\overline{p}}[D]$ is quite complex, and it is very difficult to obtain its closed-form expression.

Since we do not have a closed-form expression, we depend on noisy observations of *f*, the mean sojourn time, from the system to obtain the optimal strategy, \overline{p}^* . Here is where the deep neural network (DNN) fits in. They are state-of-the-art tools used for several learning problems, especially regression. Before we proceed with the motivation for using DNN, it is worth mentioning that several stochastic approximation algorithms, such as simultaneous perturbation stochastic approximation ([38], pp. 41–76), exist for such noisy function optimization. However, the convergence of such algorithms is prone to high variance in the gradient estimate and often leads to suboptimal results. In fact, we tried SF-SPSA ([38], pp. 77–102) in our system and observed that the algorithm leads to a suboptimal point in many cases. ReLU (rectified linear unit) based deep neural networks (DNN), on the other hand, are adept at approximating such complex functions on compact subsets, such as \mathbb{P} [39]. Particularly, it is seen that DNN can provide better generalization in function approximation, even with noisy training data [40]. Further, DNNs are also known to provide good gradient approximations for the approximated function [41]. This motivates us to use DNN to approximate $f(\overline{p})$ as $f_{\theta}(\overline{p})$, where θ is the weight parameter of the DNN. Further, the gradients required for optimization are derived using the finite difference method on $f_{\theta}(\overline{p})$. Another important feature of our algorithm is the replay memory. This idea is borrowed from the reinforcement learning setting [42]. It helps us in storing previously seen noisy function observations and using it for training the DNN in online fashion.

The replay memory and online training of the DNN are the important features of our algorithm. Online training inherently adds an importance sampling [43] feature to our algorithm, that is, we train our neural network only with samples that are more informative. This is shown to accelerate the DNN training time [43]. We see in our algorithm that this happens naturally, as training samples for the neural network come from the parameter \overline{p} update step. These samples give more information about the neighborhood of the point that the algorithm is currently in, thereby improving the confidence/variance in the descent direction. We now present our algorithm, deep assisted stochastic gradient descent, for obtaining the optimal queuing strategy.

Deep Assisted Stochastic Gradient Descent (DSGD)

Our algorithm has three steps as follows:

• Obtaining noisy observation \hat{f} of the function f at random points and storing it in replay memory, \mathbb{M}_D . This provides us with the initial training set.

.

To obtain \hat{f} for a randomly generated point \overline{p} , the system is set to follow policy \overline{p} and run until the S_{approx} services are completed. Let d_i be the sojourn time of the *i*th successfully served request in S_{approx} services. These are stored in a temporary memory \overline{D} . From d_i , $i \in [|\overline{D}|]$ compute the following:

$$\hat{f} = \frac{1}{|\overline{D}|} \sum_{i=1}^{|\overline{D}|} d_i \tag{1}$$

The point (\overline{p}, \hat{f}) is stored in \mathbb{M}_D , and \overline{D} is cleared.

• Sample a minibatch of points from \mathbb{M}_D , and uniformly randomly and train f_θ as follows:

$$\theta \leftarrow \theta - \eta_1 \nabla_{\theta} L_{f_{\theta}} \tag{2}$$

where $L_{f_{\theta}}$ is the mean square error obtained from the minibatch sampled from the replay memory, given by $L_{f_{\theta}} = \sum_{i=1}^{n} (f_{\theta}(\overline{p}_i) - \hat{f}_i)^2 / n$.

Obtain the numerical gradient of f_{θ} at the last executed point \overline{p} and perform a gradient descent as follows:

$$\overline{p} \leftarrow \mathcal{P}(\overline{p} - \eta_2 \nabla_{\overline{p}} f_\theta(\overline{p})) \tag{3}$$

Obtain the noisy observation of f at the new point. Store the new (\overline{p}, \hat{f}) to the replay memory, \mathbb{M}_D . \mathcal{P} is the projection operator that projects the input to the probability simplex as follows:

$$\mathcal{P}[r_1, r_2, r_3] = \{[r_1, r_2, r_3]\}^+ / \sum_{i=1}^3 \{r_i\}^+)$$
(4)

where the element wise operator $\{\cdot\}^+ = max\{0, \cdot\}$, and $r_i \in \mathbb{R}$, i = 1, 2, 3.

 η_1 and η_2 are learning parameters and must follow the learning rate relationships of the multi-timescale stochastic gradient descent [9] given in (17) in Section 5. The detailed algorithm is given in Algorithm 1.

Note 1: The initial training phase ($t < T_{train}$) and the explorative noise $Unif([0, \epsilon_t]^3)$ in Algorithm 1 avoid pathological zero gradients, which may stall the algorithm prematurely. Further, we use the Adam optimizer [44] in all our SGD steps for gradient annealing.

Note 2: The online learning of the DNN weights is important for adapting to a changing environment, which is experienced in practical systems. The samples in replay memory \mathbb{M}_D are collected along the descent trajectory in small steps. Thus, DNN is trained to learn the local surface/neighborhood at each step and improve the gradients as the algorithm progresses. This is importance sampling for DNN.

Section 7.2 provides the simulation results of DSGD for a multicast system with constant transmit power.

Algorithm 1 Deep assisted stochastic gradient descent (DSGD) algorithm

Input:

Multicast system in Section 2.1, Replay Memory: \mathbb{M}_D , Minibatch size: *n*, Training Time: *T_{train}*, Approximation Window: *S_{approx}*, Initialize neural network weights: θ of f_{θ} , Exploration Parameter: $\epsilon(t) \to 0, \theta, \overline{p}$ learning rates: $\eta_1(t), \eta_2(t)$ must satisfy (17), Simulation Time: T, Algorithm timeline: t, Multicast System timeline: s for t = 1 to T do if $(t < T_{train})$ then $\overline{p} \leftarrow \mathcal{P}(Unif([0,1]^3))$ else **Sample:** Minibatch *n* from \mathbb{M}_D /*Perform DNN θ update and \overline{p} parameter update as follows:*/ $\theta \leftarrow \theta - \eta_1 \nabla_{\theta} L_{f_{\theta}}$ $\overline{p} \leftarrow \mathcal{P}(\overline{p} - \eta_2 \nabla_{\overline{p}} f_\theta + Unif([0, \epsilon_t]^3))$ end run S_{approx} Multicast services with strategy \overline{p} and store $d'_i s$ in \overline{D} obtain f, as in (1) \rightarrow clear D store (\overline{p}, \hat{f}) in \mathbb{M}_D end $\overline{p}^* \leftarrow \overline{p}$ **Output:** \overline{p}^* : Optimal Queueing Strategy

4. Power Control for Multicast Queue

We now proceed to describe the power control in the Multicast setup. Adapting the transmit power based on the system and environment state under certain system constraints helps in providing the power control that may improve QoS, which is quantified by the mean user delay under stationarity. It is shown in [8] that by choosing the transmit power based on the channel gains, the system performance improves. We describe the system constraint, a power control model and the MADS power control algorithm proposed in [8] in this section. We then end this section with the Markov decision process formulation of the entire system that aids in the development of the deep reinforcement learning based power control algorithm.

4.1. Average Power Constraint

Depending on the value of H(t) and V(t) at time t, the server chooses transmit power P_t based on a power control policy $P_t = \pi(H(t), V(t))$. Choosing a good power control policy is the topic of this section. The state S_t of the system at time t is (H(t), V(t)). Let P_{S_t} be the power chosen by a policy for state S_t and let $R(S_t, P_{S_t})$ be the number of successful transmissions for the selected power P_{S_t} during the tth service.

For a fixed transmission rate *C* and for a given channel gain H(t) of users, the transmit power requirement P_{req} (from Shannon's Formula) for user *j* is (assuming file length is long enough) the following:

$$P_{req}(j, S_t) = \frac{N_g}{H_i^2(t)} (2^{C/B} - 1),$$
(5)

where *B* is the bandwidth and N_g is the Gaussian noise power at receiver *j*. Here, for simplicity, we take the ideal Shannon formula in (5), which can be easily modified to make it more realistic ([45], Chapter 14). Thus, the reward for the chosen power control policy, during the *t*th transmission, is given by the following:

$$R(S_t, P_{S_t}) = \sum_{j=1}^{L} V_{j,S_t} \, \mathbf{1}_{\{P_{S_t} > P_{req}(j,S_t)\}}(t), \tag{6}$$

where $V_{j,S_t} = 1$ if the user *j* has requested the file in service, and $V_{j,S_t} = 0$ otherwise. We now describe the mesh adaptive direct search (MADS) power control policy.

4.2. MADS Power Control Policy

The power control policy in [8] is derived from the following optimization problem,

$$\max_{\{P_1,\cdots,P_K\}}\sum_{k=1}^K q_k R_k \ s.t. \ \sum_{k=1}^K q_k P_k \le \overline{P} \ \text{and} \ P_k \ge 0, k \in [K],$$
(7)

where $[K] := \{1, \dots, K\}, \overline{P}$ is the average power constraint, K is the total number of states, P_k is the power chosen by the policy in state k, q_k is the stationary distribution of state $k \in [K]$ and are assumed to be known apriori, and R_k is the reward for state k, given as $R_k = R(S_t = k, P_t = P_k)$. This is a non-convex optimization problem since the reward in Equation (6) is a simple function (linear combination of indicators). Mesh adaptive direct search (MADS) [46] is used in [8] to solve this constrained optimization problem and obtain the power control policy. Though MADS achieves a global optimum, it is not scalable, as its computational complexity is very high.

The state space and action space of this problem can be very high, even for a moderate number of users and channel gains, e.g., a system with *L* users and *G* channel gain states has $O(2^L G^L)$ states. Therefore, in this paper, we propose a deep reinforcement learning framework. This not only provides optimal solution for a reasonably large system, but does so without knowing the arrival rates and channel gain statistics. In addition, we show via simulations that we can track an optimal solution, even when the arrival and channel gain statistics change with time.

4.3. MDP Formulation

The above system can be formulated into a finite state, action Markov decision process denoted by a tuple (S, A, *r*, **P**, γ —the state space, action space, reward, transition probability, and discount factor), where the transition probability $\mathbf{P}(S_{t+1}|S_0, P_0, ..., S_t, P_t) = \mathbf{P}(S_{t+1}|S_t, P_t)$, policy π chooses power $P_t \sim \pi(.|S_t)$ in state S_t and the instantaneous reward $r_t = R(S_t, P_t)$.

The action–value function [47] for this discounted MDP for policy π is as follows:

$$Q^{\pi}(s,a) = \mathbb{E}[\sum_{t=0}^{\infty} \gamma^{t} r_{t} | S_{0} = s, P_{0} = a].$$
(8)

where $0 < \gamma < 1$. The optimal *Q*-function, Q^* , is given by $Q^*(s, a) = \max_{\pi} Q^{\pi}(s, a)$ and satisfies the following optimality relation:

$$Q^{*}(s,a) = r(s,a) + \max_{a'} \gamma \mathbb{E}[Q^{*}(s',a')],$$
(9)

where s' is sampled with distribution $\mathbf{P}(.|s, a)$. If we know the optimal Q-function (Q^*) , we can compute the optimal policy via $\pi(s) = \arg \max Q^*(s, a)$. We know the transition

matrix of this system and hence, can compute the Q-function. However, the state space is very large, even for a small number of users, rendering the computations unfeasible. Thus, we use a parametric function approximation of the Q-function via deep neural networks and use DeepRL algorithms to obtain the optimal Q^* . Our cost function is the stationary mean sojourn time. To obtain a policy which minimizes this, we actually should be working with the average cost MDP instead of discounted MDP. However, the RL formulation for this problem is defined for the discounted case, the average case being more complicated. If we take the discount factor gamma close enough to 1, then the optimal policy obtained via the discounted problem is often close to the average case problem.

Further, to introduce the average power constraint in the MDP formulation, we look at the policies achieving the following:

$$Q^*(s,a) = \max_{\pi: C_P \le \overline{P}} Q^{\pi}(s,a)$$
(10)

where

$$C_P = \mathbb{E}[\lim_{T \to \infty} \frac{\sum_{t=0}^{T} P_t}{T}]$$
(11)

is the long term average power. We use the Lagrange method for constrained MDPs [29] to achieve the optimal policy. In this method, the instantaneous reward is modified as follows:

$$r_t = R(S_t, P_t) - \beta P_t, \tag{12}$$

where β is the Lagrange constant achieving optimal Q^* while maintaining $C_P \leq \overline{P}$. Choosing β wrongly provides the optimal policy with an average power constraint that is different from \overline{P} .

5. Deep Reinforcement Learning Based Power Control Policy

In this section, we describe the deep Q network (DQN) [11] based power control. First, we describe the DQN algorithm. We then propose a variant of DQN for constrained problems wherein we use a Lagrange multiplier to take care of the average power constraint. We use a multi-time scale stochastic gradient descent approach to also learn the Lagrange multiplier to obtain the right average power constraint. Finally, we change the learning step size from decreasing to a constant so that the optimal power control can track the time varying system statistics.

5.1. Deep Q Networks

DQN is a popular deep reinforcement learning algorithm to handle large statespace MDPs with unknown/complex dynamics $\mathbf{P}(S_{t+1}|S_t, P_t)$. The DQN is a value iteration based method, where the action-value function is approximated by a neural network. Though there are several follow-up works providing improvements over this algorithm [17,48], we use this algorithm owing to its simplicity. We show that DQN itself is able to provide us with the optimal solution and tracking. These improvements may further improve the performance in terms of sample efficiency, estimator variance, etc. Earlier attempts for combining nonlinear function approximators, such as neural networks and RL, were unsuccessful due to instabilities caused by (1) correlated training samples, (2) a drastic change in policy with small change in function approximation, and (3) correlation between the training function and approximated function [10]. The success of DQN is attributed to addressing these issues with two key ingredients of the algorithm: experience replay memory \mathbb{M} and target network, Q_{θ^*} . The replay memory stores the transitions of an MDP, specifically the tuple (S_t, P_t, r_t, S_{t+1}) . The algorithm then samples, uniformly, a random minibatch of transitions from the memory. This removes correlation between the data and smooths the data distribution change with the iteration. The algorithm has another neural network, approximating the value function, Q_{θ} . The target network and randomly sampled mini-batch from the memory $\mathbb M$ form the training set for training the Q_{θ} at every epoch. This random sampling provides i.i.d samples for performing stochastic gradient descent with loss as follows:

$$L_Q^{\pi_{\theta}} = \frac{1}{n} \sum_{j=1}^n (Y_j - Q_{\theta}(S_j, A_j))^2$$
(13)

where $Y_i = r_i + \gamma \max_{a'} Q_{\theta^*}(S_i, a')$. The iterations $\{\theta_t\}$ are given by the following:

$$\theta_{t+1} \leftarrow \theta_t - \eta_1(t) \nabla_{\theta} L_{\Omega}^{\pi_{\theta}}, \tag{14}$$

12 of 24

where $\eta_1(t)$, the step size, satisfies the following:

$$\sum_{t=0}^{\infty} \eta_1(t) = \infty, \ \sum_{t=0}^{\infty} \eta_1^2(t) < \infty, \ \eta_1(t) \ge 0.$$
(15)

The weights of the target network Q^* are held constant for T_{target} epochs, thereby controlling any drastic change in policy and reducing correlation between Q and Q^* . This can be seen as a risk minimization problem in non-parametric regression with regression function Q_{θ^*} and risk $L_Q^{\pi_{\theta_t}}$. Readers are referred to [16] for an elaborate analysis of DQN. Theorem 4.4 in [16] provides a proof of the convergence and rate of convergence, using non-parametric regression bounds, when sparse ReLU networks are used, under certain smoothness assumptions on the reward function and the dynamics.

5.2. Adaptive Constrained DQN (AC-DQN)

The DQN algorithm is meant for unconstrained optimization. Since our problem has an average power constraint of \overline{P} , we consider the instantaneous reward in (12) with a Lagrange multiplier β . The long-term constraint depends on the Lagrange multiplier and can be quite sensitive to it. Thus, we design our algorithm, AC-DQN, to learn the appropriate β . We will see later, that this enables us to further modify our algorithm to track the changing statistics of the channel gains and arrival statistics. The AC-DQN algorithm is given in Algorithm 2. Here, we use the multi-timescale SGD as in [9]. In this approach, in addition to the SGD on Q_{θ} , using a minibatch, we use a stochastic gradient descent on the Lagrange constant, β , as follows:

$$\beta_{t+1} \leftarrow \beta_t + \eta_2(t) \nabla_\beta L_P^{\pi_\theta},\tag{16}$$

where $\nabla_{\beta}L_{P}^{\pi_{\theta}} = C_{P}(S_{t}) - \overline{P}$. Since the expectation in (11) is not available to us, we take $C_{P}(S_{t}) = \sum_{i=t-T_{W}}^{t} P_{i}(S_{i})/T_{W}$, where T_{W} is the finite horizon window. Additionally, η_{1} and η_{2} are required to follow the following [9]:

$$\sum_{i=1}^{\infty} \eta_1(i) = \sum_{i=1}^{\infty} \eta_2(i) = \infty, \ \sum_{i=1}^{\infty} \eta_1^2(i) + \eta_2^2(i) < \infty, \ \frac{\eta_2(i)}{\eta_1(i)} \to 0.$$
(17)

Tracking with AC-DQN: The tracking of system statistics is essential to achieve optimal power control in a non-stationary system. In multi-time scale stochastic gradient descent, such as AC-DQN, step sizes $\eta_1(t)$ and $\eta_2(t)$ can be fixed to enable tracking. If $\eta_2 << \eta_1$, then the Lagrange multiplier changes much more slowly than the *Q*-function. Then, the two timescale theory (see, e.g., [9]) allows the Lagrange multiplier to adapt slowly to the changing system statistics but at the same time provide average power control. The solution reaches in a neighborhood of the optimal point. Although the convergence of this modified algorithm is not proved yet (even for the unconstrained DQN, convergence was proved only recently in [16]), our simulations show that the resulting algorithm tracks the optimal solution in the time varying scenario.

The time varying scenario in our setup results due to change in the request arrival statistics from the users and the changing channel gain statistics due to the motion of the users.

Algorithm 2 Adaptive constrained DQN (AC-DQN) algorithm

Input:

MDP- $(S, A, r, \mathbf{P}, \gamma)$, r as in (12), Replay Memory: \mathbb{M} , Minibatch size: n, Initialize T, T_{target} , θ , θ^* of Q_{θ} and Q_{θ^*} , Exploration Parameter: $\epsilon(t) \to 0$, Lagrange Constant: β , Value and Lagrange learning rates: $\eta_1(t)$, $\eta_2(t)$ must satisfy (17), Initialize T_W for t = 1 to T do Observe state S_t , Apply action $A_t = \pi_t(S_t) = \arg \max Q_\theta(S_t, a')$, ϵ -greedily Observe: r_t, S_{t+1} Store: $(S_t, A_t, r_t, C_P(S_t), S_{t+1})$ in \mathbb{M} **Sample:** Minibatch *n* from \mathbb{M} for i = 1 to n do $Y_i = r_i + \gamma \max_{a'} Q_{\theta^*}(S_{i+1}, a')$ end /*Perform two time-scale stochastic gradient descent as follows:*/ $\theta \leftarrow \theta - \eta_1 \nabla_{\theta} L_O^{\eta_{\theta}}$ $\beta \leftarrow \beta + \eta_2 \nabla_\beta L_P^{\pi_\theta}$ at every $t = mT_{target}, m \in \mathbb{N}^+$: update $\theta^* \leftarrow \theta$ end $\pi^* \leftarrow \pi_T, \theta^* \leftarrow \theta$ **Output:** Q_{θ^*} : Optimal Q-Function, π : Optimal Policy

6. Integrated DSGD and AC-DQN (IDA)

We are now familiar with how the multi-time scale stochastic gradient descent can be used for optimization of a stochastic system with multiple objectives. We extend this idea to learn the optimal queuing strategy while learning the optimal power control policy and simultaneously satisfying the average power constraint. Toward this, we add DSGD as a third timescale to AC-DQN. Though DSGD internally has two stochastic gradient descent steps, we consider it to be a combined third step of IDA for conceptual clarity. We present our integrated DSGD and AC-DQN (IDA) in Algorithm 3. There are four learning rates involved in the algorithm. The four learning rates should satisfy the following criteria for convergence of the algorithm [9]:

$$\sum_{i=1}^{\infty} \eta_j(i) = \infty, \ j = 1, 2, 3, 4,$$

$$\sum_{i=1}^{\infty} \sum_{j=1}^{4} \eta_j^2(i) < \infty, \ \frac{\eta_{j+1}(i)}{\eta_j(i)} \to 0, \ j = 1, 2, 3.$$
(18)

Though this criterion is required for convergence, we have seen that constant step sizes are helpful in tracking. So, we see our simulations with $\eta_1 > \eta_2 > \eta_3/T_{approx} > \eta_4/T_{approx}$.

Note 3: IDA can be used in systems with multiple objectives, e.g., a wireless network with cross layer objectives. It is important to select carefully the objective to be optimized in the slower timescale and in the faster timescale. In our setup, we run the learning steps for queuing policy (DSGD step) in a slower time scale to avoid drastic changes in the underlying MDP (of AC-DQN step).

Note 4: Step sizes in the algorithm are important hyperparameters. A good set of step sizes ensures a balance between speed and stability of the gradient descent steps. The choice is problem dependent and heuristical.

We now present the simulation results of all the algorithms presented in this paper.

Algorithm 3 Integrated DSGD and AC-DQN algorithm (IDA) Input: **DQN Input**: MDP-(\mathbb{S} , \mathbb{A} , r, \mathbf{P} , γ), r as in (12), \mathbb{M} , n, T, T_{target} , θ , θ^* of Q_{θ} , Q_{θ^*} , $\epsilon(t) \rightarrow 0, \beta, \eta_1(t) \rightarrow 0, \eta_2(t) \rightarrow 0$ and T_W are same as in Algorithm 2, **DSGD Inputs:** Replay Memory: \mathbb{M}_D , Minibatch size: n_D , Training Time: S_{train} , Approximation Window: T_{approx} , Initialize weights θ^{\dagger} of $f_{\theta^{\dagger}}$, Exploration Parameter: $\epsilon_D(s) \rightarrow 0$, Learning rates: η_i , i = 1, 2, 3, 4 satisfy, (18), System timeline: *t*, DSGD timeline: $s, s \leftarrow 0, \overline{p} \leftarrow \mathcal{P}(Unif([0, 1]^3))$ for t = 1 to T do Observe S_t , Take action A_t and store $(S_t, A_t, r_t, C_P(S_t), S_{t+1})$ in \mathbb{M} **Sample:** Minibatch *n* from \mathbb{M} as in Algorithm 2 /*AC-DQN Step: Perform two time-scale stochastic gradient descent as follows:*/ $\theta \leftarrow \theta - \eta_1 \nabla_{\theta} L_Q^{\pi_{\theta}},$ $\beta \leftarrow \beta + \eta_2 \nabla_\beta L_P^{\pi_\theta}$ at every $t = mT_{target}, m \in \mathbb{N}^+$: update $\theta^* \leftarrow \theta$ $\overline{D} \leftarrow append(sojourtime d'_{i}s)$ in current service if $t = mT_{avvrox}, m \in \mathbb{N}^+$ then if $(s < S_{train})$ then $\overline{p} \leftarrow \mathcal{P}(Unif([0,1]^3))$ else **Sample:** Minibatch n_D from \mathbb{M}_D /*DSGD Step: Perform DNN θ^{\dagger} update and \overline{p} parameter update as follows:*/ $\theta^{\dagger} \leftarrow \theta^{\dagger} - \eta_3 \nabla_{\theta^{\dagger}} L_{f_{\theta_{\star}}},$ $\overline{p} \leftarrow \mathcal{P}(\overline{p} - \eta_4 \nabla_{\overline{p}} f_{\theta^{\dagger}}(\overline{p}) + Unif([0, \epsilon_D(s)]^3))$ end obtain \hat{f} , as in (1) \rightarrow clear $\overline{D} \rightarrow$ store (\overline{p}, \hat{f}) in \mathbb{M}_D $s \leftarrow s + 1$ end $\overline{p}^* \leftarrow \overline{p}, \pi^* \leftarrow \pi_T, \theta^* \leftarrow \theta$ **Output:** \overline{p}^* : Optimal Queueing Strategy, Q_{θ^*} : Optimal Q-Function, π : Optimal Policy

7. Simulation Results and Discussion

In this section, we first present the simulation results for our DSGD algorithm. We run the multicast system with constant transmit power. We compare the performance of our DSGD queuing algorithm against each queuing strategy proposed in [8]. Next, we compare the performance of AC-DQN and MADS power control policies. We show that the deep learning algorithm, AC-DQN, indeed achieves the global optimum obtained by the MADS algorithm, but unlike MADS, is also scalable with the system size (number of users). We further demonstrate that the AC-DQN algorithm tracks the changing system dynamics and obtains the optimal policy, adaptively. Finally, we consider IDA. We show, numerically, that the algorithm achieves the optimal point obtained by both DSGD and AC-DQN. We use Python 3.8 with the Tensorflow 2.4.0/Keras package for system implementation (the system and algorithm codes are available at https://github.com/rkraghu88/SchedulingPC_IDA, accessed on 12 October 2021).

7.1. Simulation Parameters

We consider three systems with varying system configurations as follows.

7.1.1. Small User Case

Number of users L = 4, catalog size M = 100, file size F = 10 MB, transmission rate C = 10 MB/s, bandwidth B = 10 MHz, channel gains ~Uniform([0.1 0.2 0.3]) for two users with bad channel statistics and ~Uniform([0.7 0.8 0.9]) for two users with good channel

statistics. File popularity: uniform, (Zipf exponent = 0), average power constraint $\overline{P} = 7$, simulation time = 10^5 multicast transmissions.

7.1.2. Moderate User Case

System parameters: power transmit levels = 20 (1 to 50), L = 10, M = 100, F = 10 MB, C = 10 MB/s, channel gains: exponentially distributed ($\sim exp(0.1)$ for bad channel, $\sim exp(1.0)$ for good channel), R = 10 MB/s, B = 10 MHz, $\overline{P} = 7$. File popularity: Zipf distribution with Zipf exponent = 1. Simulation time: 10^5 multicast transmissions. In both the cases, we set the noise power as $N_g = 1$.

7.1.3. Large User Case

System Parameters: Same as Section 7.1.2 except, L = 20.

7.1.4. Hyperparameters

For DSGD, we consider a fully connected neural network with two hidden layers. The first layer has 32 nodes, and the second layer has 16 nodes. All layers have a ReLU activation function. $\mathbb{M}_D = 1000$, Minibatch size: $n_D = 50$, $T_{train} = 100$: Training Time, $S_{approx} = 100$: Approximation Window, Initialize weights θ of f_{θ} , $\epsilon(t) \rightarrow 0$, $\eta_1(t) = 0.01/(1+0.00001t)$, $\eta_2(t) = 0.001/(1+0.00001t(llog(t)))$.

Finally, in the IDA algorithm, we combine the parameters of both DSGD and AC-DQN. The step sizes are, however, held constant with the value of each step size at t = 0.

7.2. Optimal Queueing Using DSGD

We consider the moderate user system in Section 7.1.2 for demonstrating the performance of DSGD. We assume the widely accepted IRM traffic model with unity zipf popularity for the 100 different file requests arriving at 10 users. The server is endowed, in different simulation runs, with different queuing strategies. We compare our DSGD based queuing strategy at the server with the individual queuing strategies, mentioned in Section 2. The server transmits the files with constant transmit power $\overline{P} = 7$. We model the wireless fading to follow Rayleigh distribution. This introduces the errors in file transmissions.

We see in Figure 2a that different queuing strategies are optimal at different rates for a constant transmit power $\overline{P} = 7$ under fading. This is the typical case in practical systems. Depending on the request load, the system might need to adapt the queuing and service strategy. DSGD does precisely this. We can see in Figure 2b that the algorithm converges to the optimal mean sojourn time for the given power policy. We use a constant transmit power policy. Epochs 0 to 10^4 are the initial training phase, and the algorithm starts learning thereafter and eventually converges. The policies chosen by the algorithm for arrival rates 0.6 and 3.0 are given in Figure 3a,b, respectively. We see that for rate 3.0, the algorithm converges to the defer strategy since it has the lowest mean sojourn time for this rate (Figure 2a). For rate 0.6 however, we see that DSGD gives a mixed policy with positive probabilities to retransmit and loopback and zero probability to defer. This is because both retransmit and loopback have the same mean delay performance, and the defer strategy performs poorly. This is the case where more than one optimal solution may be available, and the algorithm may converge to one or oscillate between different optimal points as neural network training progresses. The simulations show that the DSGD algorithm adaptively chooses the best among the three queuing policies or an equivalent mixed policy for different system statistics (arrival rates).



(a) DSGD mean sojourn times vs. arrival rate.

(b) DSGD convergence of mean sojourn time.

Figure 2. DSGD performance in parameterized multicast system with constant power policy, L = 10, $\overline{P} = 7$, Zipf Popularity (Zipf exponent = 1), Rayleigh fading with mean, 0.1 and 1.0 for bad and good users, respectively.



Figure 3. Probability convergence for L = 10, $\overline{P} = 7$, Zipf popularity (Zipf exponent = 1), Rayleigh fading with mean, 0.1 and 1.0 for bad and good users, respectively.

7.3. Optimal Power Control (AC-DQN vs. MADS)

We use the system setting of the small user case, specified in Section 7.1.1, since running MADS for a higher number of users is computationally prohibitive. We use the uniform popularity profile for the file requests. We also use uniform distribution for fading. This is just for the convenience of the calculations of state probabilities, $\{q_k\}$, in MADS as done in [8]. We compare the performance of AC-DQN and MADS for this system. We demonstrate our algorithm with more realistic distribution in the next section.

We use the loopback queuing strategy for demonstrating AC-DQN. We see in subsequent sections that AC-DQN works even for other queuing strategies. To show the advantage of power control, we split the users in two equal sized groups, where one group has good channel statistics and the other bad channel statistics. We compare the performance of both the power control policies with the constant power control policy, where the transmit power P_t is fixed to $P_t = \overline{P}$. Figure 4a shows a comparison of the mean sojourn times of constant power policy, $P_t = \overline{P}$, MADS and AC-DQN. We see from Figure 4a that AC-DQN achieves the same mean sojourn time as that by MADS but is much better than the constant power policy. Additionally, from Figure 4b, we see that AC-DQN satisfies the average power constraint.



(a) Sojourn times of MADS, PCD and AC-DQN vs. arrival rate.



(b) Convergence of average power.

Figure 4. AC-DQN performance in 1-LB system with L = 4, $\overline{P} = 7$, uniform popularity, uniform fading.

7.4. AC-DQN Tracking Simulations in a Scaled Network

The AC-DQN provides similar improvements over the constant power scheme as above, even for a large user case [49] with the 1-LB queuing scheme. In this section, we show via simulations the tracking capabilities of AC-DQN for the large user case (Section 7.1.3). We demonstrate the importance of constant step sizes for η_1 and η_2 , and the inability of decaying step sizes to track the changing system statistics. We consider a system where the arrival rates change over a period of 48 h. We fix $\lambda = 1.0$ for the first 24 h. To make the learning harder for our algorithm, we change the rates abruptly every 6 h for the next 24 h as $\lambda = 0.6, 0.5, 0.4, 0.8$. This change in time period is just to illustrate the tracking ability in a more emphatic manner. This also captures the real-world scenario, where the request traffic to the base station varies with the time of the day. We fix $\overline{P} = 5$. We calculate the mean sojourn time and average power, using a moving average window of 1000 samples in size. We run the AC-DQN algorithm for this system with (1) decaying η_1 and η_2 satisfying (17) and (2) constant step sizes, $\eta_1 = 0.001$ and $\eta_2 = 0.00003$. The rest of the parameters remain the same as in the large user case. We see in Figure 5a that the AC-DQN with a constant step size almost always outperforms the decaying step size. Specifically, after the first 24 h, the delay reduction is nearly 50 percent for a constant step size. The reason for this is evident from Figure 5b,c. We see in Figure 5c that the AC-DQN with a constant step size learns the Lagrange constant throughout the simulation time, whereas the AC-DQN decaying step size is unable to learn the Lagrange constant after the first 24 h. As can be seen in Figure 5b, this affects the average power achieved by the AC-DQN with a decaying step size. While a constant step size maintains the average power constraint of P = 5, the average power achieved by the decaying step size AC-DQN drops to 4. Hence, the decaying step size AC-DQN suffers suboptimal utilization of the available power. Thus, in practical systems, only constant step-size AC-DQN is capable of adapting to the changing system statistics. The effect of fixing the learning rates is seen in the small oscillations of average power around P = 5 in Figure 5b. This is the oscillation in a small neighborhood around the optimal average power. The smaller the step size, the lesser the oscillations.



Figure 5. AC-DQN tracking performance in 1-LB system with with decaying vs. constant step sizes L = 20, $\overline{P} = 5$, Zipf(1) popularity, Rayleigh fading.

7.5. Integrated Optimal Queueing and Power Control Using IDA

We have already seen the performance of power control for 1-LB (loopback case) for a large user system. In this section, we compare the performance of AC-DQN for different queuing strategies versus the IDA performance for the moderate user case (Section 7.1.2). We use Zipf popularity and Rayleigh fading for system simulation. First, in Figure 6a, we make an observation that AC-DQN drastically improves the mean delay performance for all the strategies as compared to the constant power policy in Figure 2a. We see that our IDA algorithm is able to choose a better strategy than the baselines in terms of the mean sojourn time. The convergence of the mean sojourn time for rates 0.2 to 3.0 is shown in Figure 6b. The more important capability of this algorithm is that it converges to a better mean sojourn time while maintaining the average power constraint. Figure 6c shows the convergence of the average power to $\overline{P} = 7$ for all the rates. This is achieved by simultaneously controlling the Lagrange variable as seen in Figure 6d. A few interesting plots showing the convergence of probabilities for rates 0.8, 2.0 and 3.0 are shown in Figure 7a–c, respectively.

We see, from Figure 7a, that for arrival rate 0.8, the queuing policy converges to a mixed policy with 0.8 probability assigned to retransmit and 0.2 assigned to loopback. This policy has the same optimal mean sojourn time as achieved by the best policy, retransmit, in Figure 6a. Thus, IDA gives additional optimal points for the algorithm to choose from. From Figure 6a we see that for rate 0.2, both defer and loopback have the same performance as AC-DQN. For arrival rate 3.0 (Figure 7c), however, IDA unambiguously chooses defer as the policy since it has the lowest mean sojourn time among the baselines in Figure 6a.



Figure 6. IDA Performance in parameterized multicast system with L = 10, $\overline{P} = 7$, Zipf popularity (Zipf exponent = 1), Rayleigh fading with mean, 0.1 and 1.0 for bad and good users, respectively.



Figure 7. IDA convergence of queuing strategies for different arrival rates. L = 10, $\overline{P} = 7$, Zipf popularity (Zipf exponent = 1), Rayleigh fading with mean, 0.1 and 1.0 for bad and good users, respectively.

7.6. Tracking of User and Rate Variation Using IDA

In wireless content-centric networks, such as Netflix over 5G networks, the traffic is generally seen to start peaking in late afternoon and reach the maximum in the evening [50]. We show that power and queuing policies are tracked simultaneously via IDA in such a non-stationary setting.

We consider a system with a maximum of 100 users accessing a library of 100 files. The rest of the system parameters are as described in Section 7.1.2. The total user and rate variation mimics the real traffic as observed in [50]. The traffic starts increasing in the late afternoon and peaks in the evening. The traffic variations over a period of four and a half days are shown in Figure 8d. Except in the first 12 hours, the number of users and the request rate vary every 6 hours. It is important to note that our state formulation in AC-DQN and IDA allows for user variation in the system. The input to the neural networks has to be chosen based on the maximum number of users of 100 that the system admits. Thus, the state-space dimension is 200.

Due to a larger state-space dimension, to improve the sample efficiency of the Q learning, we make the network deeper and increase the number of nodes [39]. We scale the neural network size to three hidden layers with 256, 128, 64 nodes, and Input:200, Output:20. For a non-stationary setting, T_{approx} should be chosen appropriately. A large T_{approx} gives a bad estimate of the mean sojourn time and a small T_{approx} increases the variance. Thus, we take $T_{approx} = 40$. Similarly, to hold only relevant samples in the memory, we reduce the replay memory \mathbb{M}_D to 100 from 1000. This $\{T_{approx}, \mathbb{M}_D\}$ corresponds to a memory of ~1 h. Additionally, to keep the exploration perpetual in a non-stationary system, we fix ϵ_D to a value of 0.005.

Figure 8 compares the performance tracking of IDA with respect to AC-DQN for individual queuing schemes when the arrival rate varies over time. This demonstrates the practical applicability of IDA where the environment is non-stationary. Figure 8a shows how IDA tracks the optimal mean sojourn time as the environment changes. During the first 12 h, the IDA assigns high probability to the retransmit policy. Since the traffic is very low in first 12 h, the learning process is slow. However, as the traffic picks up in the next 12 h, learning of the queuing policy and the power control policy, is accelerated, and IDA gets closer to the optimul performance among the three queuing schemes at 18 h. From there on, the learned optimal policy and power control are tracked near optimally. This can be seen in Figure 8a,b at time intervals [18, 24], [42, 48], [66, 72], [90, 96] hours.

Quick adaptation of IDA to changing system statistics can be seen emphatically at 50 h. At 48 h, the mean sojourn time spikes up (Figure 8a), due to change in the system statistics (Figure 8d). We deduce that the stochastic policy learned until 48 h is not optimal for the statistics of the system immediately after 48 h. Thus the f_{θ} learned in the DSGD step starts changing rapidly after 50 h, due to the small size of the replay memory, \mathbb{M}_D . This induces rapid change in the descent step of parameter p, leading to a drastically different policy. We also note that such a spike in the mean sojourn time occurs at times 68 h and 96 h. However, these spikes are not large enough to warrant a drastic policy change.

We have simulated an environment which has abrupt variations. In the real scenario, the changes are much smoother. The algorithm can only do better in such a scenario. We also observe in Figure 8c that the IDA meets the power constraint. It does so by learning the optimal Lagrange multiplier corresponding to the instantaneous environment.

It is also important to mention that IDA is a stochastic algorithm performing nonconvex optimization of a non-stationary system. The frequency of IDA getting stuck at a poor local optimum can, however, be controlled by appropriately tuning the hyperparameters, such as the DNN size, activation, step size, type of exploration noise, etc.



Time (Hours) (d) Total Arrival Rate and User Variation

60

80

Figure 8. Tracking of optimal performance by IDA in a non-stationary environment, where total arrival rate to the base station/server varies over time.

0

100

7.7. Discussion

Mean Sojourn Time (sec)

Action Probability

Average Power

8

6 4

2 0

0

20

40

Total Arrival Rate

We see from the simulations that the novel deep learning techniques, such as DSGD and AC-DQN, can achieve optimal performance while providing scalability with system size. Our two-timescale approach, AC-DQN, extends DeepRL algorithms, such as DQN, to systems with constrained control. It can be extended to systems with multiple constraints also. In such systems, each constraint is associated with a Lagrange multiplier.

For a stationary system, it is enough that the step sizes satisfy multi-timescale criteria similar to (17) (see [9]). However, if AC-DQN is used in systems with changing system statistics, the step sizes are kept constant. Choosing the step sizes is a trade-off between the tolerance of the constraint and the required algorithmic agility to track the system changes. We have also demonstrated how IDA achieves the optimal queuing strategy among the baselines while obtaining the power control for such complex multicast systems. It is shown that deep neural networks, when appropriately used, can provide scalable control for large wireless networks, achieving several cross-layer objectives. IDA also tracks the optimal performance in a large non-stationary system with varying number of users.

8. Conclusions

This paper has considered a multicast downlink in a single hop wireless network. Fading of different links to users causes significant reduction in the performance of the system. Appropriate change in the queuing policies and power control can mitigate most of the losses. However, simultaneously obtaining adaptive queuing and power control for large systems is computationally very hard. We first developed a novel DNN assisted stochastic gradient descent algorithm to achieve optimality of the system to provide a lower mean sojourn time in a parameterized multicast system. Next, we showed that, using deep reinforcement learning, we can obtain optimal power control, online, even when the system statistics are unknown. We used a recently developed version of Q learning, the deep Q network, to learn the Q-function of the system via function approximation. Furthermore, we modified the algorithm to satisfy our constraints and also to make the optimal policy track the time varying system statistics. Finally, we proposed a novel deep multi-time scale algorithm which achieves the cross-layer optimization of queuing and power control, simultaneously. We showed that IDA also performs well in a large system with a non-stationary environment.

One interesting extension of this work would be developing an algorithm that could potentially provide a better state-action-dependent queuing strategy. Another future work could possibly include the caches at the user nodes and learning the optimal caching policy along with the power control using DeepRL. Future works may also consider applying IDA to multiple-base-station scenarios for interference mitigation. Further, extension of the approach to a multi-agent approach with decentralized execution as in [51] is an important research direction.

Author Contributions: Formal analysis, R.R. and V.S.; Funding acquisition, V.S.; Methodology, R.R., M.P. and V.S.; Software, R.R.; Supervision, V.A. and V.S.; Writing—original draft, R.R.; Writing—review & editing, V.A. and V.S. All authors have read and agreed to the published version of the manuscript.

Funding: V.A. acknowledges support from National Science Foundation under grant CNS-1618335 and CISCO Systems, Inc.

Data Availability Statement: The system and algorithm codes are available at https://github.com/ rkraghu88/SchedulingPC_IDA (accessed on 12 October 2021).

Conflicts of Interest: The authors declare no conflict of interest.

References

- Cisco. Cisco Visual Networking Index: Global Mobile Data Traffic Forecast Update 2016–2021 White Paper. Available online: http://www.czechmarketplace.cz/news/cisco-visual-networking-index-global-mobile-data-traffic-forecast-update-20 16-2021-white-paper (accessed on 4 December 2017).
- Cha, M.; Kwak, H.; Rodriguez, P.; Ahn, Y.Y.; Moon, S. I Tube, You Tube, Everybody Tubes: Analyzing the World's Largest User Generated Content Video System. In Proceedings of the 7th ACM SIGCOMM Conference on Internet Measurement, San Diego, CA, USA, 24–26 October 2007; pp. 1–14.
- Moghadam, N.; Li, H. Improving queue stability in wireless multicast with network coding. In Proceedings of the 2015 IEEE International Conference on Communications (ICC), London, UK, 8–12 June 2015; pp. 3382–3387.
- Cogill, R.; Shrader, B. Queue length analysis for multicast: Limits of performance and achievable queue length with random linear coding. In Proceedings of the 2009 47th Annual Allerton Conference on Communication, Control, and Computing (Allerton), Monticello, IL, USA, 30 September–2 October 2009; pp. 462–468.
- 5. Maddah-Ali, M.A.; Niesen, U. Fundamental limits of caching. IEEE Trans. Inf. Theory 2014, 60, 2856–2867. [CrossRef]
- 6. Rezaei, F.; Khalaj, B.H. Stability, rate, and delay analysis of single bottleneck caching networks. *IEEE Trans. Commun.* **2016**, 64, 300–313. [CrossRef]

- Panju, M.; Raghu, R.; Sharma, V.; Aggarwal, V.; Ramachandran, R. Queueing Theoretic Models for Uncoded and Coded Multicast Wireless Networks with Caches. *IEEE Trans. Wirel. Commun.* 2021. [CrossRef]
- Panju, M.; Raghu, R.; Agarwal, V.; Sharma, V.; Ramachandran, R. Queueing theoretic models for multicasting under fading. In Proceedings of the 2019 IEEE Wireless Communications and Networking Conference (WCNC), Marrakesh, Morocco, 15–18 April 2019.
- 9. Borkar, V. Stochastic Approximation: A Dynamical Systems Viewpoint; Cambridge University Press: Cambridge, UK, 2008.
- 10. Li, Y. Deep Reinforcement Learning. arXiv 2017, arXiv:1701.07274.
- 11. Mnih, V.; Kavukcuoglu, K.; Silver, D.; Rusu, A.A.; Veness, J.; Bellemare, M.G.; Graves, A.; Riedmiller, M.; Fidjeland, A.K.; Ostrovski, G.; et al. Human-level control through deep reinforcement learning. *Nature* **2015**, *518*, 529–533. [CrossRef]
- 12. Schulman, J.; Levine, S.; Abbeel, P.; Jordan, M.; Moritz, P. Trust region policy optimization. In Proceedings of the International Conference on Machine Learning, ICML, Lille, France, 7–9 July 2015; pp. 1889–1897.
- 13. Schulman, J.; Wolski, F.; Dhariwal, P.; Radford, A.; Klimov, O. Proximal Policy Optimization Algorithms. *arXiv* 2017, arXiv:1707.06347.
- 14. Mnih, V.; Kavukcuoglu, K.; Silver, D.; Graves, A.; Antonoglou, I.; Wierstra, D.; Riedmiller, M. Playing Atari with Deep Reinforcement Learning. *arXiv* 2013, arXiv:1312.5602.
- Silver, D.; Hubert, T.; Schrittwieser, J.; Antonoglou, I.; Lai, M.; Guez, A.; Lanctot, M.; Sifre, L.; Kumaran, D.; Graepel, T.; et al. A general reinforcement learning algorithm that masters chess, shogi, and Go through self-play. *Science* 2018, 362, 1140–1144. [CrossRef]
- 16. Fan, J.; Wang, Z.; Xie, Y.; Yang, Z. A Theoretical Analysis of Deep Q-Learning. In Proceedings of the 2nd Annual Conference on Learning for Dynamics and Control, PMLR, Berkeley, CA, USA, 11–12 June 2020.
- 17. Van Hasselt, H.; Guez, A.; Silver, D. Deep reinforcement learning with double q-learning. In Proceedings of the Thirtieth AAAI Conference on Artificial Intelligence, Phoenix, AZ, USA, 12–17 February 2016.
- 18. Moghadam, N.; Li, H.; Zeng, H.; Liu, L. Lyapunov Scheduling and Optimization in Network Coded Wireless Multicast Network. *IEEE Trans. Veh. Technol.* **2018**, *67*, 5135–5145. [CrossRef]
- Moghadam, N.; Zhang, G.; Li, H. Simplified Optimal Scheduling (SOS) for Network Coded Wireless Multicast. In Proceedings of the 2018 IEEE 88th Vehicular Technology Conference (VTC-Fall), Chicago, IL, USA, 27–30 August 2018; pp. 1–5.
- 20. Zhou, Y.; Zhao, Z.; Qi, C.; Li, R.; Louet, Y.; Palicot, J.; Zhang, H. Multicast scheduling for delay-energy trade-off under bursty request arrivals in cellular networks. *IET Commun.* **2019**, *13*, 1696–1701. [CrossRef]
- 21. Poularakis, K.; Iosifidis, G.; Sourlas, V.; Tassiulas, L. Exploiting Caching and Multicast for 5G Wireless Networks. *IEEE Trans. Wirel. Commun.* **2016**, *15*, 2995–3007. [CrossRef]
- 22. Jindal, N.; Goldsmith, A. Capacity and optimal power allocation for fading broadcast channels with minimum rates. *IEEE Trans. Inf. Theory* **2003**, *49*, 2895–2909. [CrossRef]
- Wang, K.; Chiasserini, C.F.; Rao, R.R.; Proakis, J.G. A distributed joint scheduling and power control algorithm for multicasting in wireless ad hoc networks. In Proceedings of the IEEE International Conference on Communications, ICC'03, Anchorage, AK, USA, 11–15 May 2003; Volume 1, pp. 725–731.
- 24. Zhou, B.; Cui, Y.; Tao, M. Optimal dynamic multicast scheduling for cache-enabled content-centric wireless networks. *IEEE Trans. Commun.* **2017**, *65*, 2956–2970. [CrossRef]
- 25. Ye, H.; Li, G.Y.; Juang, B.F. Deep Reinforcement Learning Based Resource Allocation for V2V Communications. *IEEE Trans. Veh. Technol.* **2019**, *68*, 3163–3173. [CrossRef]
- 26. Nasir, Y.S.; Guo, D. Multi-Agent Deep Reinforcement Learning for Dynamic Power Allocation in Wireless Networks. *arXiv* 2018, arXiv:1808.00490.
- 27. Zhang, Z.; Chen, H.; Hua, M.; Li, C.; Huang, Y.; Yang, L. Double Coded Caching in Ultra Dense Networks: Caching and Multicast Scheduling via Deep Reinforcement Learning. *IEEE Trans. Comm.* **2020**, *68*, 1071–1086. [CrossRef]
- Mao, Q.; Hu, F.; Hao, Q. Deep Learning for Intelligent Wireless Networks: A Comprehensive Survey. IEEE Commun. Surv. Tutor. 2018, 20, 2595–2621. [CrossRef]
- 29. Altman, E. Constrained Markov Decision Processes; CRC Press: Boca Raton, FL, USA, 1999.
- 30. Gattami, A.; Bai, Q.; Aggarwal, V. Reinforcement Learning for Constrained Markov Decision Processes. In Proceedings of the International Conference on Artificial Intelligence and Statistics, San Diego, CA, USA, 13–15 April 2021.
- 31. Ding, D.; Wei, X.; Yang, Z.; Wang, Z.; Jovanović, M.R. Provably Efficient Safe Exploration via Primal-Dual Policy Optimization. In Proceedings of the International Conference on Artificial Intelligence and Statistics, San Diego, CA, USA, 13–15 April 2021.
- 32. Agarwal, M.; Bai, Q.; Aggarwal, V. Concave Utility Reinforcement Learning with Zero-Constraint Violations. *arXiv* 2021, arXiv:2109.05439.
- 33. Bai, Q.; Bedi, A.S.; Agarwal, M.; Koppel, A.; Aggarwal, V. Achieving Zero Constraint Violation for Constrained Reinforcement Learning via Primal-Dual Approach. *arXiv* 2021, arXiv:2109.06332.
- Achiam, J.; Held, D.; Tamar, A.; Abbeel, P. Constrained policy optimization. In Proceedings of the International Conference on Machine Learning, Sydney, Australia, 6–11 August 2017; pp. 22–31.
- 35. Tessler, C.; Mankowitz, D.J.; Mannor, S. Reward Constrained Policy Optimization. arXiv 2018, arXiv:1805.11074.
- 36. Yu, M.; Yang, Z.; Kolar, M.; Wang, Z. Convergent Policy Optimization for Safe Reinforcement Learning. *Adv. Neural Inf. Process. Syst.* **2019**, *32*, 3127–3139.

- 37. Sadeghi, P.; Kennedy, R.A.; Rapajic, P.B.; Shams, R. Finite-state Markov modeling of fading channels—A survey of principles and applications. *IEEE Signal Process. Mag.* 2008, 25, 57–80. [CrossRef]
- 38. Bhatnagar, S.; Prasad, H.; Prashanth, L. Stochastic Recursive Algorithms for Optimization: Simultaneous Perturbation Methods; Springer: London, UK, 2013.
- Hanin, B. Universal Function Approximation by Deep Neural Nets with Bounded Width and ReLU Activations. *Mathematics* 2019, 7, 992. [CrossRef]
- 40. Rolnick, D.; Veit, A.; Belongie, S.; Shavit, N. Deep Learning is Robust to Massive Label Noise. arXiv 2017, arXiv:1705.10694.
- 41. Nguyen-Thien, T.; Tran-Cong, T. Approximation of functions and their derivatives: A neural network implementation with applications. *Appl. Math. Model.* **1999**, 23, 687–704. [CrossRef]
- 42. Lin, L.J. Self-improving reactive agents based on reinforcement learning, planning and teaching. *Mach. Learn.* **1992**, *8*, 293–321. [CrossRef]
- Katharopoulos, A.; Fleuret, F. Not All Samples Are Created Equal: Deep Learning with Importance Sampling. In Proceedings of the 35th International Conference on Machine Learning (ICML 2018), Stockholm, Sweden, 10–15 July 2018; Volume 80, pp. 2525–2534.
- 44. Kingma, D.P.; Ba, J. Adam: A Method for Stochastic Optimization. arXiv 2014, arXiv:1412.6980.
- 45. Proakis, J.; Salehi, M. Digital Communications, 5th ed.; McGraw-Hill: New York, NY, USA, 2008.
- 46. Audet, C.; Dennis, J.E., Jr. Mesh adaptive direct search algorithms for constrained optimization. *SIAM J. Optim.* **2006**, *17*, 188–217. [CrossRef]
- 47. Puterman, M.L. *Markov Decision Processes: Discrete Stochastic Dynamic Programming*, 1st ed.; J. Wiley & Sons, Inc.: Hoboken, NJ, USA, 1994.
- Lillicrap, T.P.; Hunt, J.J.; Pritzel, A.; Heess, N.; Erez, T.; Tassa, Y.; Silver, D.; Wierstra, D. Continuous control with deep reinforcement learning. *arXiv* 2015, arXiv:1509.02971.
- Raghu, R.; Upadhyaya, P.; Panju, M.; Agarwal, V.; Sharma, V. Deep Reinforcement Learning Based Power Control for Wireless Multicast Systems. In Proceedings of the 2019 57th Annual Allerton Conference on Communication, Control, and Computing (Allerton), Monticello, IL, USA, 24–27 September 2019.
- 50. Laterman, M.; Arlitt, M.; Williamson, C. A campus-level view of Netflix and Twitch: Characterization and performance implications. In Proceedings of the 2017 International Symposium on Performance Evaluation of Computer and Telecommunication Systems (SPECTS), Seattle, WA, USA, 9–12 July 2017; pp. 1–8. doi: 10.23919/SPECTS.2017.8046774. [CrossRef]
- Liu, C.; Geng, N.; Aggarwal, V.; Lan, T.; Yang, Y.; Xu, M. CMIX: Deep Multi-agent Reinforcement Learning with Peak and Average Constraints. In *Joint European Conference on Machine Learning and Knowledge Discovery in Databases*; Springer: Berlin/Heidelberg, Germany, 2021; pp. 157–173.